Author’s response to reviews

Title: Trajectories of clinical and parenting outcomes following admission to an inpatient motherbaby unit

Authors:

Nicole Reilly (nicole.reilly@newcastle.edu.au)

Elloise Brake (e.brake@unsw.edu.au)

Nancy Briggs (nancy.briggs@unsw.edu.au)

Marie-Paule Austin (m.austin@unsw.edu.au)

Version: 2 Date: 14 Oct 2019

Author’s response to reviews:

Author responses to Editor requests

1. Please confirm whether informed consent obtained from all participants was written or verbal, and clearly state this in your Methods and Ethics approval and consent to participate sections. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation. Furthermore, please clarify if the patients consented to analysis of their medical records and if any further permission from the hospital was required.

Amended as requested, and now reads: ‘….Women who agreed to participate gave written consent …’ and ‘…Participants also consented to their inpatient information being used for research purposes…’

2. Please make sure the abstract contains only the following headings: Background, Methods, Results, Conclusions.

Amended as requested.

3. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

‘Conclusions’ section added as requested.

4. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. If your manuscript uses identifying images or other personal/clinical details, please include a statement of
consent to publish from the patient, or in the case of minors, the patients’ guardians. If this is not applicable to your manuscript please state “Not Applicable” in this section.

Amended as requested, and now reads: ‘..Not applicable ..’

5. Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file. Please remove any figures from the main text of the manuscript.

Each figure has been uploaded as a single file, as requested.

6. Please provide figure titles and legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Added as requested.

7. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. This includes simple titles like 'Figure 1'.

Figure titles have been removed, as requested

8. Currently, you cite a table 2 but have not uploaded a table 2. Please either upload a table 2, or remove this citation.

Table 2 has been uploaded as requested

9. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Added as requested.

10. Please rename your 'Ethics approval' section to 'Ethics approval and consent to participate'

Amended as requested

11. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Uploaded as requested